Author's response to reviews

Title: "That never would have occurred to me": a qualitative study of medical students' views of a cultural competence curriculum

Authors:

Johanna Shapiro (jfshapir@uci.edu)
Desiree Lie (dalie@uci.edu)
David Gutierrez (dgutier@uci.edu)
Gabriella Zhuang (gzhuang@uci.edu)

Version: 4 Date: 19 May 2006

Author's response to reviews:

Many thanks again to the reviewers who have worked so diligently with us on this article. Reviewer 1 did not suggest any changes in the manuscript. Reviewer 2 continued to express concern about the term "theoretical triangulation." In deference to this concern, we have substituted the suggested term "investigator triangulation," and have removed the following sentence discussing theoretical triangulation. No other changes or revisions were suggested.